Author's response to reviews

Title: The frequent BRCA1 mutation 1135insA has multiple origins: a haplotype study in different populations

Authors:

Teresa M Rudkin (teresa@mrs.mni.mcgill.ca)
Nancy Hamel (nancy.hamel@mail.mcgill.ca)
Maria Galvez (maria.galvez@mail.mcgill.ca)
Frans Hogervorst (f.hogervorst@nki.nl)
Johan JP Gille (jjg.gille@vumc.nl)
Pal Moller (pal.moller@medisin.uio.no)
Jaran Apold (jaran.apold@helse-bergen.no)
William D Foulkes (william.foulkes@staff.mcgill.ca)

Version: 6 Date: 15 February 2006

Author's response to reviews:

- The full stop was removed from the Title of the manuscript
- In order to establish coherence between the manuscript and the submission system, a comma was added to the following sentence: “In centres where targeted mutation testing is performed, exclusively…”
- On the author list, the accents were put on Pal Moller’s name on the submission system